Author’s response to reviews

Title: Impact of postoperative major complications on long-term survival after radical resection of gastric cancer

Authors:

Peng Yuan (peng8264503@126.com)
Zhouqiao Wu (wuzhouqiao@hotmail.com)
Ziyu Li (ligregory369@hotmail.com)
Zhaode Bu (buzhaode@gmail.com)
Aiwen Wu (wuaw@sina.com)
Xiaojiang Wu (wu.xiaojiang@hotmail.com)
Lianhai Zhang (zlhzlh@hotmail.com)
Jinyao Shi (yuan_peng@outlook.com)
Jiafu Ji (jijiafu@hsc.pku.edu.cn)

Version: 2 Date: 27 Apr 2019

Author’s response to reviews:

1. Please ensure you have included your tables and figures within your submission. We note that currently your tables and figures are missing.

   author: I had included my tables and figures within my submission

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

   author: I had provide a list of all the abbreviations which be placed before the Declarations section
3. Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

author:This version was clean